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Weekly Summary 03 July – 10 July
6.3E18 POT, 85.6% Uptime

• Horn continues to operate flawlessly after 
power supply repairs three weeks ago.

• Major Downtime:

– 7 hrs: Booster RF and MI12 cooling repairs 
(July 5).

– 7 hrs: ACNET-DAQ (IRM data) problems.  
Beam data recoverable (July 8).

• MI-12 cooling problems ongoing, working on a 
permanent fix.
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One Week MB Proton/Hr Summary

2.5E16 p/hr

03 July 10 July

Protons/Spill

Protons/Hr

NUMI Off, running at 9 turns. Lots of protons!

9 turns



This Weeks Plan 

• Continue to see improvements in Booster 
output.  Steady at 9 turns, started going 
to 10 turns.  Close to >4E12 ppp, which is 
preferred mode of operation.

• Continue taking quality anti-neutrino 
physics data.


